\begin{frame}
\onslide<2>{\begin{itemize}
		\item{Pumping fee reduced extraction by 33\% \citep{smith2017}}
		\item{Economically distinct aspects of nuclear markets}
		\item{Unique economic cost considerations}
\end{itemize}}
\end{frame}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%\begin{frame}{Location of Subdistrict 1 wells\begin{spacing}{0}
%\tiny \citep{coloradodivisionofwaterresourcesStructures2024}\end{spacing}}
%\centering
%	\includegraphics[width=\textwidth]{Chapter_I/images/SLV_WELL_MAP.png}
%\end{frame}
%%%
\begin{frame}{Location of Subdistrict 1 wells}
\centering
	\includegraphics[width=\textwidth]{Chapter_I/images/SLV_WELL_MAP.png}
\begin{spacing}{0}
\hfill\tiny Map creation tool from \citep{coloradodivisionofwaterresourcesStructures2024}\end{spacing}
\end{frame}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{frame}{San Luis Valley Agriculture}
	\begin{columns}
		\begin{column}{0.57\textwidth}
			\begin{itemize}
				\onslide<2->{\item{Agriculture began as early as 1630 \citep{hearne1988}}}
				\onslide<3->{\item{350,000 irrigated acres by 1890 \citep{carlson1973}}}
			\end{itemize}
		\end{column}
		%%%%%
		\begin{column}{0.43\textwidth}
			\includegraphics[width=\textwidth]{Chapter_I/images/farmers.jpg}
			\newline
			\tiny \hfill San Luis Valley potato harvest \citep{rothsteinUntitledPhotoPossibly1939}
			\includegraphics[width=\textwidth]{Chapter_I/images/potatoe.jpg}
			\newline
			\tiny \hfill San Luis Valley potato field \citep{krakel2024}
		\end{column}
\end{columns}

\end{frame}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{frame}
\begin{figure}%
	\centering
	\includegraphics[width=0.8\textwidth]{Chapter_I/images/SLV_Map.png}
	\caption{San Luis Valley Parcel Map}\label{MAP}%
\end{figure}%
\end{frame}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{frame}{Contemporary Accounts}
	\begin{block}{Doug Shriver, potato and wheat farmer}
	``I feel it is important that we handle our own problems, and this is an opportunity for us to fix those problems ourselves rather than have the State fix them for us. It really is the right thing to do, because we caused the problem." \citep{heide2005}\end{block}
%\begin{quotation}``I think it is inevitable that ground in the San Luis Valley is going to be fallowed whether by subdistricts or CREP or well regulation or simply running out of adequate supplies of water....I do care about the valley, and think I am in a position to understand what we are facing whether it feels like ‘Chicken Little’ or not. [We have] to try to come to some resolution without a wreck." \citep{heide2005}\end{quotation}
%\hspace{\parindent}\hspace{\parindent}\hspace{\parindent}--Ray Wright, President, Rio Grande Water Conservation District 
%\newline
	\begin{block}{Ray Wright, President, Rio Grande Conservations District}``They’ve pumped the bottom out of the barrel. And we are trying to get a groundwater management subdistrict up there, and it is like pulling teeth...In other areas, people are very interested in trying to put something together...A lot of people will say, ‘Oh, they will never come in here and regulate wells.’ And Hal Simpson [the State Engineer] has told them several times, ‘If you don’t do something, I’m going to be here'." \citep{ralphcurtis2005}\end{block}
\end{frame}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{frame}
	\input{Chapter_I/tables/SBD_CONT_BAL_TBL.tex}
\end{frame}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{frame}[plain]
	\only<1>{
	\input{Chapter_I/tables/All_Models.tex}
}
	\only<2>{
	\input{Chapter_I/tables/All_Models_Small.tex}
}
\end{frame}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{frame}
\begin{figure}%
	\centering
	\includegraphics[width=\textwidth]{Chapter_I/images/Iplot_Raw.png}
	\caption{Event study of Subdistrict One formation}\label{REG:EVENTW}%
\end{figure}%
\end{frame}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{frame}
\begin{figure}%
	\centering
	\includegraphics[width=\textwidth]{Chapter_I/images/Iplot_Weighted.png}
	\caption{Event study of Subdistrict One formation with parcel weighting}\label{REG:EVENTW}%
\end{figure}%
\end{frame}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

